Author’s response to reviews

Title: Mazabraud's Syndrome: First case of fibrous dysplasia of bone associated with soft-tissue myxomas as well as an intra-osseous myxoma: a case report.

Authors:

Wybren M.D. van der Wal (w.vanderwal@maartenskliniek.nl)
Halil M.D. Unal (h.unal@zrt.nl)
Jacky M.D. de Rooy (j.derooy@rad.umcn.nl)
Uta M.D. Flucke (u.flucke@pathol.umcn.nl)
Rene M.D. Ph.D. Veth (r.veth@orthop.umcn.nl)

Version: 3 Date: 16 December 2010

Author’s response to reviews:

We hereby submit the new, revised version of our case report about a patient with Mazabraud's Syndrome.

All authors took another, critical look at the case report, and some adjustments were made, which are highlighted.

Indeed there was a mistake in the correlation between figures, their legends and the text. This was corrected.

There was a sentence in the article about an osseous cyst. It was a mistake that this sentence was still there. Due to multiple revisions and removal of text because it was too long, we removed the following text about the multiple laesions which were found 3 years after the last surgery:
"two proximally anterior, one proximally posterior and two halfway the femur shaft in the vastus intermedius muscle. Another lesion was seen at the posterolateral tibia plateau"

This piece of text is about the exact localisation of the 5 new soft tissue laesions. Next to this 5 laesions another laesion was seen of which we think this was a synovial cyst with degenerative origin. We think this finding is not relevant for the case report and that's why we deleted it now (which we should have done in the first revision already, it was confusing now). If you think it is relevant to mention we can include it in the text.

It was a mistake which we did not see with the last revision.

Hopefully the case report can be accepted now.